Department of Mathematical Sciences Spring 2002 Carnegie Mellon University

21-112 Calculus II

Announcement of Test #1

Test #1 will be administered in lecture on Friday, February 8, 2002. This is a closed-book and closed-note exam. Calculators are not permitted.

Test #1 will cover Sections 5.1, 5.2, 5.3, 5.5, 5.6, 5.7 and 5.8. From section 5.7, you are responsible only for the material on consumer’s and producer’s surplus. From Section 5.8, you are responsible only for the material on finding the volume of a solid of revolution and finding the average value of a function.